Author’s response to reviews

Title: The development of a brief screener for autism using item response theory

Authors:

Caroline Mårland (caroline.marland@neuro.se)

Gitta Lubke (glubke@nd.edu)

Alessio Degl' Innocenti (alessio.deglinnocenti@vregion.se)

Maria Råstam (maria.rastam@med.lu.se)

Christopher Gillberg (christopher.gillberg@gnc.gu.se)

Thomas Nilsson (thomas.nilsson@neuro.gu.se)

Sebastian Lundström (sebastian.lundstrom@neuro.gu.se)

Version: 2 Date: 15 Oct 2019

Author’s response to reviews:

Dear editor,

Please find the attached manuscript with the required formatting changes.

Sincerely,

Sebastian Lundström